Statistics 1xx-Level Help Sessions

August 31, 2001

In order to better serve the students enrolled in the 1xx-series courses, ISDS offers help sessions Sunday through Thursday from 4 to 6 PM in room 025 Old Chemistry building. Teaching Assistants will help students grasp key concepts via guiding them through their course work. Additionally, TAs are available to assist with any questions regarding class assignments.

In assisting with assignments, however, it is the students’ responsibility to have attempted the work before attending help session hours. TAs are not there to do students’ assignments, rather will direct them on the right path to understanding statistical concepts and ideas.

While students from any 1xx-level class may attend any help session hours, it may be beneficial to come on days when a TA from their particular class is available. That way, the TA assisting the student will be aware of the requirements, techniques, and intricacies of that specific course and its assignments. Nonetheless, TAs from any 1xx-level course are available to help you.

We’re here to help YOU, so feel free to come visit help sessions anytime!

Thus, the following is a schedule of the TAs who will be working during help session hours.

Time Sun. Mon. Tues. Wed. Thur.

4-5 Jinseoq Philippe Christine Merrill Laura Leanna or Steve Tracy Enrique Tom Binnur Alister Xia Rosy Binnur Tracy 4-4:30

5-6 Jinseoq Philippe Christine Merrill Laura Leanna or Steve Tracy 4-5:30 Enrique Tom Binnur Xia Rosy Allister Leanna